**Author’s response to reviews**

**Title:** Antepartum urinary tract infection and postpartum depression in Taiwan - a nationwide population-based study

**Authors:**

Jui-Ming Liu (mento1218@gmail.com)

Feng-Hsiang Chiu (kru24tw@163.com)

Yueh-Ping Liu (dtemer14@gmail.com)

Shu-Pin Chen (vic102535@yahoo.com.tw)

Hsun-Hao Chan (j505065@gmail.com)

Jing-Jung Yang (jjyang0827@gmail.com)

Fung-Wei Chang (doc30666@gmail.com)

REN-JUN HSU (hsurnai@gmail.com)

**Version:** 4  **Date:** 08 Feb 2018

**Author’s response to reviews:**

Editor Comments:

Please address the following final editorial requests:

1) Figures and legends:

- Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

- Figure files should contain only the image/graphic, as well as any associated keys/annotations. Please remove titles/legends from within the figure files.

- Note that each figure should be uploaded as separate file in the submission system, and fit on a single A4 page in portrait format.

Ans. Thank you for your comments. We have modified the figure you mentioned before. Please see new figure 1.
2) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Ans. Thank you for your comments. We have modified the manuscript you mentioned. Please see new manuscript.

BMC Pregnancy and Childbirth operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Hilary Davies (Reviewer 2): Thank you for sending the revised version, I am happy with the revisions made.

Ans, Thank you for your comments.